Author's response to reviews

Title: Australian women's use of complementary and alternative medicines to enhance fertility: exploring the experiences of women and practitioners

Authors:

Jo-Anne Rayner (j.rayner@latrobe.edu.au)
Helen L McLachlan (h.mclachlan@latrobe.edu.au)
Della A Forster (d.forster@latrobe.edu.au)
Rhian Cramer (arn_22_@hotmail.com)

Version: 4 Date: 9 December 2009

Author's response to reviews:

Iratxe Puebla
Senior Editor,
BMC-series journals
9th December 2009

Dear Iratxe

Re: MS: 6035653812982822
‘Australian women's use of complementary and alternative medicines (CAM) to enhance fertility: exploring the experiences of women and practitioners’.

Thank you for the reviews of the above paper. Please find below a point by point response to the formatting changes required by the journal. These changes have been made.

Thankyou again for publishing this paper,

Yours sincerely

Dr Jo-Anne Rayner

1. Major revisions (we require the author to make these changes): A ‘Conclusions’ section has been included after the ‘Discussion’.

2. Typography: The manuscript has been checked for typographical errors and all corrections have been made.